Author's response to reviews

Title: Treatment outcomes of patients who migrate for HIV care in the province of British Columbia, Canada, from 2003 to 2012: a retrospective cohort study

Authors:

Viviane D. Lima (vlima@cfenet.ubc.ca)
Nichola Goldberg (ngolde@gmail.com)
Lillian Lourenço (llourenco@cfenet.ubc.ca)
William Chau (wchau@cfenet.ubc.ca)
Robert S. Hogg (bobhogg@cfenet.ubc.ca)
Silvia Guillemi (sguillemi@cfenet.ubc.ca)
Rolando Barrios (rbarrios@cfenet.ubc.ca)
Julio S.G. Montaner (jmontaner@cfenet.ubc.ca)

Version: 2
Date: 16 March 2015

Author's response to reviews:

Re: MS: 1592298790163504
Treatment outcomes of patients who migrate for HIV care in the province of British Columbia, Canada, from 2003 to 2012: a retrospective cohort study
Viviane D. Lima, Nichola Goldberg, Lillian Lourenço, William Chau, Robert S. Hogg, Silvia Guillemi, Rolando Barrios and Julio S.G. Montaner
BMC Medicine
Dear Dr Lima

Many thanks for your recent submission to BMC Medicine (referenced above).
My colleagues and I have carefully considered your manuscript and I'm very sorry to say it seems to us that it would not be appropriate for BMC Medicine. It is our feeling that the manuscript does not represent a sufficiently profound step forward in understanding to be of broad interest to researchers outside the field.

I believe BMC Health Services Research would be appropriate for your research, please review the scope for yourself
http://www.biomedcentral.com/bmchealthservres/about/#aimsscope.

The transfer of your manuscript does not imply that it will be accepted and the Editors are free to reject the manuscript if it is not suitable.

Accept the transfer to BMC Health Services Research by logging on to 'My Manuscripts' http://www.biomedcentral.com/bmcmed/my/manuscripts.

You will also find further details including the new Article Processing Charge and peer-review policy.
Thank you for your interest in BMC Medicine. I do hope that you will consider an alternative BioMed Central journal for your manuscript and publish with BMC Health Services Research.

Similarly to BMC Medicine, BMC Health Services Research also operates using an open peer-review system. Please refer to http://www.biomedcentral.com/bmchealthservres/about#publication for further details on the peer-review policy of BMC Health Services Research, or if you have any questions, please contact editorial@biomedcentral.com.

Thank you for your interest in BMC Medicine.

Best wishes,

Alessandro

Mr Alessandro Recchioni
Assistant Editor, BMC Medicine

Tel: +44 (0) 20 3192 2000
e-mail: bmcmedicineeditorial@biomedcentral.com

Web: http://www.biomedcentral.com/